Author’s response to reviews

Title: Determinants of Academic Performance in Children with Sickle Cell Anaemia

Authors:

Osita Ezenwosu Dr (docsita31@yahoo.com)
Ifeoma Emodi Prof (ifeoma_emodi@yahoo.com)
Anthony Ikefuna Dr (nnaikefuna@yahoo.com)
Barth Chukwu Dr (barth.chukwu@unn.edu.ng)
Chidiebere D.I Osuorah Dr (chidi.osuorah@gmail.com)

Version: 2 Date: 27 May 2013

Author’s response to reviews:

COVER LETTER

1. Title Page:

Title page has been included at the front of the revised manuscript with names, institutions, countries and e-mail addresses of all authors as well as the full correspondence of the submitting author.

2. Competing interests:

Competing interests section has been placed after the conclusions. However, there is no competing interest in relation to the manuscript.

3. Authors’ information:

Authors’ contributions section has been included after competing interests section. Also, relevant information about the authors has been included under ‘authors’ information’ section after authors’ contribution section.

4. Acknowledgement:

The Committee and Ministry who contributed towards the study by giving approval and clearance respectively were acknowledged after authors’ information.

5. Conclusions:

The main conclusions of the research as well as the explanation of their importance and relevance have been included under the discussion.

6. References:

This has been formatted according to the instructions for authors.
Abstract:

Abstract ‘introduction’ was changed to ‘background’ and ‘objective’ incorporated into the background. Results of the ‘mean of overall academic scores’, ‘mean DAPQ’ and ‘mean number of days of absence’ for subjects and controls were removed in the abstract. This is in order to limit the abstract to the instructed ‘not more than 350 words’.